

GOPEN ACCESS

Citation: Laugesen K, Veres K, Hernandez-Diaz S, Chiu Y-H, Oberg AS, Hsu J, et al. (2023) Overweight or obesity in children born after assisted reproductive technologies in Denmark: A population-based cohort study. PLoS Med 20(12): e1004324. https://doi.org/10.1371/journal. pmed.1004324

Academic Editor: Sarah J. Stock, The University of Edinburgh Usher Institute of Population Health Sciences and Informatics, UNITED KINGDOM

Received: April 24, 2023

Accepted: November 16, 2023

Published: December 19, 2023

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pmed.1004324

Copyright: © 2023 Laugesen et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data cannot be shared publicly because of Danish legislation. Data

RESEARCH ARTICLE

Overweight or obesity in children born after assisted reproductive technologies in Denmark: A population-based cohort study

Kristina Laugesen^{1*}, Katalin Veres¹, Sonia Hernandez-Diaz², Yu-Han Chiu^{2,3}, Anna Sara Oberg^{2,4}, John Hsu^{5,6,7}, Paolo Rinaudo⁸, Mandy Spaan⁹, Flora van Leeuwen⁹, Henrik Toft Sørensen¹

Department of Clinical Epidemiology, Aarhus University Hospital and Aarhus University, Aarhus, Denmark,
Department of Epidemiology, Harvard T.H. Chan School of Public Health, Boston, Massachusetts, United States of America, 3 Department of Public Health Sciences, Penn State College of Medicine, Hershey,
Pennsylvania, United States of America, 4 Department of Medical Epidemiology and Biostatistics, Karolinska Institutet, Stockholm, Sweden, 5 Mongan Institute, Massachusetts General Hospital, Boston,
Massachusetts, United States of America, 6 Department of Medicine, Harvard Medical School, Boston,
Massachusetts, United States of America, 7 Department of Debaterics Gynecology and Reproductive Sciences, University of California, San Francisco, California, United States of America, 9 Department of Epidemiology, Netherlands Cancer Institute, Amsterdam, the Netherlands

* Kristina.laugesen@clin.au.dk

Abstract

Background

The association between assisted reproductive technologies (ARTs) and the body mass index (BMI) of children remains controversial. Confounding by morbidity and other factors associated with parental infertility may have biased studies comparing children born after ART with children born after no treatment. We investigated the associations between different fertility treatments and BMI in children at age 5 to 8 years, adjusting for and stratifying by causes of parental infertility.

Methods and findings

This Danish cohort study included 327,301 children born between 2007 and 2012 (51% males, median age at follow-up 7 years). Of these, 13,675 were born after ART, 7,728 were born after ovulation induction with or without intrauterine insemination [OI/IUI], and 305,898 were born after no fertility treatments. Using the International Obesity Task Force (IOTF) standards, we defined overweight (BMI \geq IOTF-25) and obesity (BMI \geq IOTF-30). We compared children born after ART versus OI/IUI; intracytoplasmic sperm injection (ICSI) versus conventional in vitro fertilization (IVF); and frozen-thawed versus fresh embryo transfer and estimated crude and adjusted prevalences of children with overweight or obesity at age 5 to 8 years, prevalence odds ratios (PORs), and differences in mean BMI z-scores. Adjustment was performed using stabilized inverse probability of treatment weights, including parity, year of conception, parental causes of infertility, age, educational level, comorbidities, maternal country of origin, BMI, and smoking as covariates. The crude prevalence of obesity

are available through application to Danish health authorities (Statistics Denmark and the Danish Health Data Authority). The Danish Health Data Authority: www.sundhedsdatastyrelsen.dk, email kontakt@sundhedsdata.dk. Statistics Denmark: www.dst.dk, email dst@dst.dk.

Funding: This work was supported by a grant from NIH R01HD088393 received by JH. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: ART, assisted reproductive technology; ATC, Anatomical Therapeutic Chemical; BMI, body mass index; CI, confidence interval; ICSI, intracytoplasmic sperm injection; IOTF, International Obesity Task Force; IPTW, inverse probability of treatment weighting; IUI, intrauterine insemination; IVF, in vitro fertilization; OI, ovulation induction; PCOS, polycystic ovarian syndrome; POR, prevalence odds ratio; SD, standardized difference. was 1.9% in children born after ART, 2.0% in those born after Ol/IUI, and 2.7% in those born after no fertility treatment. After adjustment, children born after ART and Ol/IUI had the same prevalence of being overweight (11%; POR 1.00, 95% confidence interval [CI] 0.91 to 1.11; p = 0.95) or obese (1.9%; POR 1.01, 95% CI 0.79 to 1.29; p = 0.94). Comparison of ICSI with conventional IVF yielded similar pattern (POR 0.95, 95% CI 0.83 to 1.07; p = 0.39 for overweight and POR 1.16, 95% CI 0.84 to 1.61; p = 0.36 for obesity). Obesity was more prevalent after frozen-thawed (2.7%) than fresh embryo transfer (1.8%) (POR 1.54, 95% CI 1.09 to 2.17; p = 0.01). The associations between fertility treatments and BMI were only modestly different in subgroups defined by the cause of infertility. Study limitations include potential residual confounding, restriction to live births, and lack of detailed technical information about the IVF procedures.

Conclusions

We found no association with BMI at age 5 to 8 years when comparing ART versus OI/IUI or when comparing ICSI versus conventional IVF. However, use of frozen-thawed embryo transfer was associated with a 1.5-fold increased risk of obesity compared to fresh embryo transfer. Despite an elevated relative risk, the absolute risk difference was low.

Author summary

Why was this study done?

- The number of children born after in vitro fertilization (IVF) is increasing.
- Information about long-term health of these children is limited, and the association between IVF and body mass index (BMI) of the children remains controversial.
- Various procedures, such as type of fertilization and embryo transfer, and underlying causes of infertility may influence associations.

What did the researchers do and find?

- The researchers used Danish national registries to conduct a cohort study on the association between different fertility treatments and BMI in children at age 5 to 8 years, taking into account underlying causes of infertility.
- Children born after IVF had similar BMI compared with those born after ovulation induction (with or without intrauterine insemination) and compared with those born after no fertility treatment.
- Children born after frozen-thawed embryo transfer had higher BMI and higher prevalence of obesity compared with those born after fresh embryo transfer, compared with those born after ovulation induction (with or without intrauterine insemination), and compared with those born after no fertility treatment. Findings were consistent across subgroups defined by underlying cause of infertility.

What do these findings mean?

- Our study provides reassuring results for couples who seek help to conceive.
- Reasons behind our findings for frozen-thawed embryo transfer need further investigation.
- Longer follow-up and assessment of other outcomes related to metabolic health is needed.
- Study limitations include potential residual confounding, inclusion of live births only, and lack of detailed technical information about the IVF procedures.

Introduction

Assisted reproductive technologies (ARTs), which include manipulation ex vivo of gametes and embryo like in vitro fertilization (IVF) or intracytoplasmic sperm injection (ICSI), have led to the birth of millions of children over the past decades [1,2]. Despite the increase in live births after these fertility treatments, information about long-term health consequences remains limited.

Examination of long-term outcomes has been challenging in part because of the need for substantial follow-up time, inclusion of large numbers of children born after treatment, and acquisition of granular clinical information about both parents and children. Further, it has been difficult to separate effects of treatment from underlying causes of infertility, i.e., indication for treatment. Although some studies have linked ART to adverse cardiometabolic health in children, including obesity [3–9], the association between ART and body mass index (BMI) of children remains controversial [3–6,10–21]. Divergent findings likely are explained by differences in study design, including different exposure definitions (varying ART procedures), reference groups (children born to parents with or without infertility), and approaches to controlling confounding. Recent studies have found an association between ART and lower BMI in childhood and early adolescence compared to being born after no fertility treatment, while the association attenuated or slightly reversed in late adolescence [20,21]. Associations persisted when children born after ART were compared to children of parents with infertility who did not receive treatment [20,21]. Comparison cohorts of children with parents with no need for reproductive assistance are helpful for contextualization, but confounding from underlying parental infertility remains an issue. Associations are best examined by comparing children born by parents with infertility requiring reproductive assistance. Because infertility itself is clinically heterogeneous, information about specific causes of infertility might be needed (but is often lacking), as it remains unknown if such causes confound or modify associations. Further, specific ART procedures might have different impacts on children. For instance, frozenthawed embryo transfer is associated with a higher risk of being born large for gestational age, whereas fresh embryo transfer is associated with a higher risk of being born small for gestational age [6,15,22,23]. Also, studies indicate an association between frozen-thawed embryo transfer and higher BMI compared with fresh embryo transfer in childhood [3,6,15,20,21].

In this population-based cohort study based on national registry data, we aimed (1) to compare BMI among children aged 5 to 8 years born after ART versus ovulation induction with or without intrauterine insemination (OI/IUI); (2) to compare BMI after specific ART procedures, such as ICSI versus conventional IVF or frozen-thawed versus fresh embryo transfer; and (3) to evaluate potential effect modification by underlying causes of infertility or biologic sex of the child.

Materials and methods

This study is reported as per the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) guideline (<u>S1 STROBE</u> Checklist). The original analysis plan is shown in <u>S1 Text</u>. Some amendments were performed post hoc, including use of inverse probability of treatment weights for control of confounding and inclusion of additional potential confounders. Further, 5 extra sensitivity analyses were performed to assess robustness of our findings.

Setting and data sources

This cohort study was conducted in Denmark (which has \approx 65,000 births annually). The Danish healthcare system provides tax-supported health services to all residents, guaranteeing access to healthcare free of charge [24]. A unique personal identity number (the civil registration number) is assigned to all Danish residents at birth or upon immigration. The civil registration number is used in all registries, allowing accurate and unambiguous linkage. In Denmark, fertility treatment is paid by the public healthcare system (for the first child only) with some copayment for medical drugs. Currently, the system covers up to 3 rounds of OI/ IUI plus 3 rounds of ART. Treatments are paid by couples when they wish to conceive a second or additional child (except if using frozen embryos from prior cycles), when more than 3 ART cycles are required to conceive, or when couples seek faster access than available from the public healthcare system. We used the following data sources.

- The Danish Medical Birth Register contains information on all births in Denmark since 1973. The information is collected by the midwife or physician overseeing the delivery and includes the civil registration number of the parents and infant, as well as characteristics of the pregnancy, the delivery, and the newborn [24]. The register does not include time to pregnancy.
- The Danish Fertility Treatment Registry contains information on all ART cycles since 1994 and on all OI/IUI cycles since 2006 conducted at public and private fertility clinics. ART covers conventional IVF and ICSI. The type of embryo preparation is also specified, such as fresh or frozen-thawed embryo transfer [25].
- The Danish National Children's Database records information on all children living in Denmark since 2011 and includes variables such as height, weight, and BMI categories. At age 5 to 8 years, children enroll in primary school and attend checkups performed by health nurses. All schools in Denmark report to the database, but approximately 0.1% of all Danish children do not attend primary school (they are homeschooled or do not attend any educational offer) [26]. Moreover, children attend a routine checkup at their general practitioner at age 5, and this is also reported to the database.
- The Danish National Patient Registry contains information on all hospital inpatient admissions since 1977. Outpatient clinic contacts and emergency department visits were added in 1995. For each inpatient, outpatient clinic, and emergency department discharge, 1 primary diagnosis and optional secondary diagnoses are recorded according to the International Classification of Diseases (ICD, 8th Revision between 1977 and 1993 [ICD-8] and 10th Revision [ICD-10] thereafter) [27].

- The Danish National Prescription Registry contains information on all prescriptions filled in community pharmacies since 1995. Filled prescriptions are coded according to the Anatomical Therapeutic Chemical (ATC) Classification System [28].
- The Danish social and demographic registries contain information on income and highest educational level of completed education.
- The Danish Civil Registration System contains information on date of birth, sex, and vital status for the entire Danish population since 1968. The system is updated daily [29].

Study cohort

The Danish Medical Birth Register was used to identify members of the study cohort [24]. For this study, we identified 378,346 births in Denmark between January 1, 2007 and December 31, 2012. Stillbirths or those with duplicate civil registration numbers (1,580 [0.4%]) were excluded. A further 49,465 (13%) children were excluded who did not have routine anthropometric measurements at age 5 to 8 years recorded in the Danish National Children's Database. Thus, the final study cohort consisted of 327,301 children (Fig 1). Lack of routine anthropometric evaluation at age 5 to 8 years did not differ according to exposure or cause of infertility. Lack of BMI measurement was consistently 13% among children born after ART overall, conventional IVF, ICSI, frozen-thawed embryo transfer, fresh embryo transfer, or OI/IUI, and among children from the general population born after no fertility treatment. In addition, age at anthropometric evaluation was similar across groups (median age 7 years and interquartile range 6.6 to 7.6 years) (S2 Text).

Assisted reproductive technologies and comparison cohorts

We established exposure and comparison cohorts through linkage between the Danish Medical Birth Register and the Danish Fertility Treatment Registry [25,30]. We assessed associations for ART versus OI/IUI, ICSI versus conventional IVF, and frozen-thawed versus fresh embryo transfer. Additionally, we compared frozen-thawed embryo transfer versus OI/IUI and fresh embryo transfer versus OI/IUI. For context and to compare with previous literature, we also assessed associations for children born after ART versus children from the general population born after no fertility treatment. This latter comparison reflects the combination of the treatment and the treatment indication (i.e., infertility) and not solely the treatment.

Overweight, obesity, and body mass index

We assessed BMI at age 5 to 8 years using data recorded in the Danish National Children's Database. Our analysis was based on the International Obesity Task Force (IOTF) BMI definitions for overweight (BMI \geq IOTF-25) and obesity (BMI \geq IOTF-30) to categorize the children [31,32]. The IOTF age- and sex-specific cutoffs link BMI values at 18 years to child centiles based on data from 6 countries (Brazil, Great Britain, Hongkong, the Netherlands, Singapore, and the United States) fitted using the least mean squares method [32]. We further analysed BMI as a continous variable by using World Health Organization 2007 reference data and macro to calculate BMI z-scores [33].

Covariates

For parents with children born after fertility treatment, we identified causes of infertility, including any female factor; ovulation disorders such as polycystic ovarian syndrome (PCOS); tubal factors; uterine or cervical factors such as endometriosis; nonspecific female factor; any



Fig 1. Study cohort. Notes: Abbreviations: ART, assisted reproductive technologies; cIVF, conventional in vitro fertilization; ICSI, intracytoplasmic sperm injection; IUI, intrauterine insemination; OI, ovulation induction. *Lack of routine anthopometric evaluation did not differ according to exposure or cause of infertility. Lack of anthropometric measurement was consistently 13% among children conceived with ART overall, conventional IVF, ICSI, frozen-thawed embryo transfer, fresh embryo transfer, or OI/IUI, and among children from the general population born after no fertility treatment. Further, children with and without anthropometric measurements had similar parental and birth characteristics (S2 Text).

https://doi.org/10.1371/journal.pmed.1004324.g001

male factor; and idiopathic causes of infertility. Most of this information was obtained from the Fertility Treatment Registry and supplemented with data from the Danish National Patient Registry (<u>S3 Text</u>). For the entire study population, we retrieved information from the Medical Birth Register on year of conception, parity, maternal and paternal age at conception, maternal BMI at pregnancy start, and maternal smoking status. As a measure of socioeconomic status, we obtained information from Danish social and demographic registries on the highest maternal and paternal educational level at conception (low [primary and lower secondary education], medium [upper secondary education or academy professional degree], or high [university education at the bachelor's degree level or higher]). From the Civil Registration System, we retrieved information on maternal country of origin (Nordic, European except for the Nordic countries, or non-European). Diabetes (types 1 or 2), hyperlipidaemia or use of lipid-lowering drugs, and hypertension or use of antihypertensive drugs were identified from drug prescriptions or hospital records (inpatient and outpatient) in the Danish National Patient Registry or from National Prescription Registry records at any time before conception (S3 Text) [24]. Paternal BMI was recorded in the Fertility Treatment Registry starting in 2010. In context of potential mediation, we obtained information on gestational age at birth in weeks (<28, 28 to 31, 32 to 36, \geq 37), birth weight, large for gestational age (defined as birth weight > the 90th percentile for infants of the same gestational age, sex, and birth year), small for gestational age (defined as birth weight < the 10th percentile for infants of the same gestational age, sex, and birth year), and singleton/multiple pregnancy.

Statistical analysis

Among children born after fertility treatment, we described parental causes of infertility and treatment choices. We provided additional parental and birth characteristics among all children (i.e., born after fertility treatment or not). We further calculated the crude prevalence of overweight and obesity at age 5 to 8 years in children born after fertility treatment and in children from the general population born after no fertility treatment.

We compared children born after ART versus OI/IUI. Among children born after ART, we examined type of fertilization method, comparing children born after ICSI versus conventional IVF. We then focused on type of embryo transfer, comparing children born after frozen-thawed versus fresh embryo transfer. Additionally, we compared each type of embryo transfer to OI/IUI. We identified the following potential confounders from the literature: parental causes of infertility [34], maternal and paternal age at conception (restricted cubic splines with 3 knots) [35], maternal and paternal highest educational level at conception [36], maternal country of origin [37], maternal BMI (restricted cubic splines with 3 knots) [38], maternal smoking status [39], maternal and paternal hyperlipidemia or use of lipid-modifying drugs [40], maternal and paternal hypertension or use of antihypertensive drugs [41], presence/absence of diabetes (I or II) [42], parity [43], and year of conception [44]. To adjust for confounding, we used stabilized inverse probability of treatment weighting (IPTW) including the following steps [45]: For each child and each comparison, we used the listed potential confounders to compute a propensity score (i.e., the predicted probability of being exposed) using a logistic regression model and then graphically displayed the distribution of propensity scores across the various exposure and comparison cohorts (S1 Fig). We then used the propensity score to reweight the population by stabilized IPTW and truncated the extreme weights (the 1st and 99th percentiles) [45]. This approach created a population in which the treatment assignment was independent of the observed confounders, as these were balanced across the comparison groups. We assessed confounder balance before and after IPTW using standardized differences (SDs), with balance defined as a SD \leq 0.15. Covariates occurring postconception (birth weight, etc.) might reside on the causal pathway and were therefore not adjusted for. Using ordinal logistic regression with robust variance (generalized estimating equations), we computed crude and adjusted (weighted) prevalence odds ratios (PORs) of overweight and obesity to estimate the average treatment effect in the population undergoing fertility treatment. To estimate crude and adjusted (weighted) differences in mean BMI z-scores, we used linear regression with robust variance (generalized estimating equations). Generalized estimating equations were used to account for family clustering (i.e., 1 woman could give birth to more than 1 child). We performed subgroup analyses by cause of infertility (if allowed by sample size) and sex of the child [46]. Propensity scores and weights were recalculated within each subgroup.

We conducted multiple sensitivity analyses to assess the robustness of our findings. First, to minimize confounding by treatment history, we considered only children born in the first

ART cycle. In a second sensitivity analysis, we included the previous number of OI/IUI and ART cycles in the IPTW. These 2 types of sensitivity analyses were applicable for comparisons of ART versus OI/IUI and ICSI versus conventional IVF. Because treatments involve both the mechanism of insemination (e.g., conventional IVF or ICSI) and the type of embryo transfer (e.g., fresh or frozen), we used restriction to enable single comparisons. For example, in a third analysis, we compared frozen-thawed versus fresh embryo transfer but restricted the cohort to children of parents undergoing conventional IVF only. Fourth, to enable comparison of ICSI versus conventional IVF not mediated through embryo transfer methods, we restricted the cohort to concern fresh embryo transfers only. Fifth, we separately assessed frozen-thawed embryo transfer in modified natural cycle (using hCG trigger only) and programmed cycle (estrogen and progesterone). Sixth, to evaluate the sensitivity of estimates to incomplete adjustment for confounding associated with parental factors, we added paternal BMI to the IPTW. This analysis was performed in a subpopulation of children born in 2011 and 2012 for ART versus OI/IUI in order to obtain a sufficient number of outcomes.

Last, in supplemental analyses, we compared children born after ART versus children from the general population born after no fertility treatment. For these comparisons, we used standardized morbidity ratio weighting instead of IPTW to estimate the average effects of treatment and treatment indication combined among the exposed.

Statistical analyses were conducted using SAS version 9.4.

Ethics statement

This study was approved by the Danish Data Protection Agency (Record number: 2016-051-000001, serial number 1804). According to Danish legislation, informed consent or approval from an ethics committee is not required for registry-based studies.

Results

Parental causes of infertility and specific fertility treatments

Among the 327,301 children in our study cohort, 21,403 (6.5%) were born by parents with infertility using OI/IUI (n = 7,728) or ART (n = 13,675) (Fig 1). Causes of infertility varied among parents: ovulation disorders (n = 4,221 children of which 58.9% were born after ART); tubal factors (n = 3,222, 89.3% born after ART); cervical or uterine factors (n = 2,557, 79.7% born after ART); male factors (n = 13,097, 73.6% born after ART), and idiopathic causes of infertility (n = 789, 26.1% born after ART). The type of fertility treatment differed by cause of infertility (Table 1).

Parental and birth characteristics of children born after fertility treatment and no fertility treatment

Parental characteristics, such as age, educational level, maternal country of origin, comorbidities, and lifestyle, were similar for children born after ART versus OI/IUI, ICSI versus conventional IVF, and frozen-thawed versus fresh embryo transfer (Table 2). As expected, parental characteristics differed between children born after fertility treatment versus children born after no fertility treatment. Maternal and paternal age and educational level at conception were higher in children born after fertility treatment compared to children born after no fertility treatment (Table 2). Maternal smoking was less frequent among children born after ART (5.6%) and OI/IUI (5.7%) compared with children born after no fertility treatment (12.8%). Comorbidities, such as type 2 diabetes, hyperlipidemia/use of lipid-modifying drugs, and hypertension/use of antihypertensive drugs, were more frequent among parents in children

	N (%)									
	Female factor (any)	Ovulation disorders	Tubal factor	Cervical or uterine factor	Nonspecific female factor	Male factor	Idiopathic			
All	17,538 (100)	4,221 (100)	3,222 (100)	2,557 (100)	3,808 (100)	13,097 (100)	789 (100)			
OI/IUI	6,224 (35.5)	1,736 (40.1)	346 (10.7)	518 (20.3)	1,108 (29.1)	3,453 (26.4)	583 (74.0)			
ART	11,314 (64.5)	2,485 (58.9)	2,876 (89.3)	2,039 (79.7)	2,700 (70.9)	9,644 (73.6)	206 (26.1)			
cIVF	6,073 (34.6)	1,349 (32.0)	1,958 (60.8)	1,273 (49.8)	1,604 (42.1)	3,630 (27.7)	136 (17.2)			
ICSI	4,379 (25.0)	962 (22.8)	682 (21.2)	651 (25.5)	907 (23.8)	5,414 (41.3)	70 (8.9)			
Fresh embryo transfer	9,704 (55.3)	2,165 (51.3)	2,425 (75.3)	1,776 (69.5)	2,318 (60.9)	8,382 (64.0)	187 (2.4)			
Frozen-thawed embryo transfer	1,426 (8.1)	262 (6.2)	418 (13.0)	234 (9.2)	321 (8.4)	1,163 (8.9)	17 (2.2)			

Table 1. Parental causes of infertility* and fertility treatment procedures among 21,403 children born to parents who received fertility treatment.

*Causes of infertility are not mutually exclusive.

ART, assisted reproductive technologies; cIVF, conventional in vitro fertilization; ICSI, Intracytoplasmic sperm injection; IUI, intrauterine insemination; OI, ovulation induction.

https://doi.org/10.1371/journal.pmed.1004324.t001

born after ART and OI/IUI compared with children born after no fertility treatment (Table 2). Median maternal BMI was 23 kg/m² in all cohorts, although fewer mothers with children born after ART were obese (9.2%) compared with children born after OI/IUI (13.2%) or after no fertility treatment (11.6%) (Table 2).

The strongest predictors of overweight and obesity in children were maternal overweight or obesity, maternal country of origin, and low maternal and paternal educational levels (S4 Text).

Childbirth characteristics are shown in Table 2. The likelihood of being born small for gestional age was higher in children born after ART (13.9%) or OI/IUI (13.4%) compared with children born after no fertility treatment (9.2%) (Table 2). The prevalence was similar for ICSI (13.6%) versus conventional IVF (14.5%), but not for frozen-thawed (9.4%) versus fresh embryo transfer (14.4%). Conversely, being born large for gestional age was less prevalent in children born after ART (7.2%) or after OI/IUI (8.0%), compared with children from the general population (10.2%). With respect to specific types of ART, the likelihood of being large for gestational age was the same for ICSI (7.3%) versus conventional IVF (6.5%), but higher with frozen-thawed (11.4%) rather than fresh embryo transfer (6.6%).

Crude prevalences of overweight and obesity among children age 5 to 8 years

Median age at follow-up was 7 years for all exposure and comparison cohorts. The crude prevalences of overweight were 10% in children born after ART, 11% in children born after OI/ IUI, and 10% in children born after no fertility treatment (Fig 2). For obesity, the prevelances were 1.9%, 2.0%, and 2.7%, respectively (Fig 2). In children born after ART, the crude prevalence of obesity depended on type of fertilization (2.0% for ICSI and 1.6% for conventional IVF) (Fig 3) and type of embryo transfer (1.8% for fresh embryo transfer and 2.7% for frozenthawed embryo transfer) (Fig 4).

		N (%)									
		Chi	ldren born a	fter ART	Children born after OI/IUI	Children born after no fertility treatment					
	All ART	cIVF	ICSI	Fresh embryo	Frozen embryo						
Total	13,675 (100)	6,530 (100)	6,175 (100)	11,791 (100)	1,695 (100)	7,728 (100)	305,898 (100)				
Year of conception											
2006–2009	8,589 (62.8)	4,001 (61.3)	3,620 (58.6)	7,535 (63.9)	945 (55.8)	4,433 (57.4)	209,881 (68.6)				
2010-2012	5,086 (37.2)	2,529 (38.7)	2,555 (41.4)	4,256 (36.1)	750 (44.2)	3,295 (42.6)	96,017 (31.4)				
Previous number of OI/IUI cycles											
0	8,748 (64.0)	3,386 (51.9)	4,652 (75.3)	7,504 (63.6)	1,099 (64.8)	2,702 (35.0)	NA				
1	911 (6.7)	476 (7.3)	378 (6.1)	781 (6.6)	119 (7.0)	1,896 (24.5)	NA				
2	808 (5.9)	493 (7.5)	267 (4.3)	711 (6.0)	91 (5.4)	1,315 (17.0)	NA				
<u>≥</u> 3	3,208 (23.5)	2,175 (33.3)	878 (14.2)	2,795 (23.7)	386 (22.8)	1,815 (23.5)	NA				
Previous number of ART cycles											
0	4,689 (34.3)	2,618 (40.1)	1,802 (29.2)	4,490 (38.1)	127 (7.5)	7,245 (93.8)	NA				
1	3,188 (23.3)	1,433 (21.9)	1,403 (22.7)	2,499 (21.2)	669 (39.5)	162 (2.1)	NA				
2	2,290 (16.7)	1,003 (15.4)	1,131 (18.3)	1,907 (16.2)	368 (21.7)	113 (1.5)	NA				
<u>≥</u> 3	3,508 (25.7)	1,476 (22.6)	1,839 (29.8)	2,895 (24.6)	531 (31.3)	208 (2.7)	NA				
Oocyte donation	187 (1.4)	70 (1.1)	102 (1.7)	NA	NA	NA	NA				
Sperm donation	547 (4.0)	479 (7.3)	62 (1.0)	493 (4.2)	53 (3.1)	2,187 (28.3)	NA				
Maternal characteristics											
Age at conception (median, IQR)	33 (30–36)	34 (31– 37)	33 (30- 36)	33 (30–36)	33 (31–36)	33 (30–36)	30 (27–34)				
Parity											
0	8,573 (62.7)	4,093 (62.7)	3,873 (62.7)	7,542 (64.0)	905 (53.4)	4,932 (63.8)	129,856 (42.5)				
≥1	4,721 (34.5)	2,259 (34.6)	2,110 (34.2)	3,903 (33.1)	762 (45.0)	2,616 (33.9)	164,471 (53.8)				
Missing	381 (2.8)	178 (2.7)	192 (3.1)	346 (2.9)	28 (1.7)	180 (2.3)	11,571 (3.8)				
Highest educational level											
Low	1,010 (7.4)	493 (7.5)	444 (7.2)	864 (7.3)	124 (7.3)	624 (8.1)	51,656 (16.9)				
Medium	5,530 (40.4)	2,567 (39.3)	2,570 (41.6)	4,776 (40.5)	678 (40.0)	2,952 (38.2)	124,487 (40.7)				
High	6,894 (50.4)	3,345 (51.2)	3,054 (49.5)	5,932 (50.3)	875 (51.6)	4,053 (52.4)	118,737 (38.8)				
Missing	241 (1.8)	125 (1.9)	107 (1.7)	219 (1.9)	18 (1.1)	99 (1.3)	11,018 (3.6)				
Country of origin											
Nordic	12,474 (91.2)	5,993 (91.8)	5,605 (90.8)	10,760 (91.3)	1,545 (91.2)	7,100 (91.9)	259,342 (84.8)				
European (except Nordic countries)	457 (3.3)	208 (3.2)	211 (3.4)	392 (3.3)	56 (3.3)	232 (3.0)	15,278 (5.0)				
Non-European	637 (4.7)	282 (4.3)	307 (5.0)	544 (4.6)	83 (4.9)	327 (4.2)	26,743 (8.7)				
Missing/unknown	107 (0.8)	47 (0.7)	52 (0.8)	95 (0.8)	11 (0.6)	69 (0.9)	4,535 (1.5)				

Table 2. Baseline characteristics of 327,301 children born in Denmark during 2007–2012 who underwent routine anthropometric evaluation at age 5-8 years.

(Continued)

Table 2. (Continued)

	N (%)									
		Chi	ldren born a	fter ART		Children born after OI/IUI	Children born after no fertility treatment			
	All ART	cIVF	ICSI	Fresh embryo	Frozen embryo					
BMI at start of pregnancy (kg/m ²) (median IQR)	23 (21–26)	23 (21– 26)	23 (21– 27)	23 (21–26)	23 (21–26)	23 (21–27)	23 (21–26)			
<18.5	448 (3.3)	238 (3.6)	181 (2.9)	380 (3.2)	64 (3.8)	247 (3.2)	12,050 (3.9)			
18.5–24	8,441 (61.7)	4,202 (64.3)	3,633 (58.8)	7,260 (61.6)	1,067 (62.9)	4,441 (57.5)	176,549 (57.7)			
25-29	2,854 (20.9)	1,231 (18.9)	1,409 (22.8)	2,446 (20.7)	365 (21.5)	1,646 (21.3)	59,509 (19.5)			
30-34	968 (7.1)	397 (6.1)	519 (8.4)	847 (7.2)	103 (6.1)	702 (9.1)	22,999 (7.5)			
≥35*	291 (2.1)	149 (2.3)	126 (2.0)	262 (2.2)	26 (1.5)	315 (4.1)	12,463 (4.1)			
Missing/ implausible	673 (4.9)	313 (4.8)	307 (5.0)	596 (5.1)	70 (4.1)	377 (4.9)	22,328 (7.3)			
Smoking during pregnancy										
Yes	759 (5.6)	397 (6.1)	303 (4.9)	660 (5.6)	90 (5.3)	441 (5.7)	39,234 (12.8)			
Missing	399 (2.9)	191 (2.9)	188 (3.0)	360 (3.1)	28 (1.7)	194 (2.5)	12,985 (4.2)			
Diabetes (type I, II)	619 (4.5)	315 (4.8)	272 (4.4)	536 (4.5)	69 (4.1)	574 (7.4)	5,886 (1.9)			
Type I diabetes	71 (0.5)	33 (0.5)	37 (0.6)	62 (0.5)	8 (0.5)	36 (0.5)	1,297 (0.4)			
Type 2 diabetes	602 (4.0)	282 (4.3)	235 (3.8)	474 (4.0)	61 (3.6)	538 (6.9)	4,589 (1.5)			
Hyperlipidaemia/ Lipid-modifying drugs	99 (0.7)	51 (0.8)	39 (0.6)	82 (0.7)	17 (1.0)	67 (0.9)	1,542 (0.5)			
Hypertension/ antihypertensive drugs	1,681 (12.3)	811 (12.4)	765 (12.4)	1,445 (12.3)	197 (11.6)	1,080 (14.0)	32,480 (10.6)			
Paternal characteristics										
Age at conception, (median, IQR)	35 (32–39)	35 (32- 39)	35 (32- 39)	35 (32–39)	35 (32–39)	34 (31–38)	32 (29–36)			
Highest educational level										
Low	1,374 (10.0)	605 (9.3)	651 (10.5)	1,161 (9.8)	190 (11.2)	745 (9.6)	54,796 (17.9)			
Medium	7,067 (51.7)	3,189 (48.8)	3,350 (54.3)	6,114 (51.9)	860 (50.7)	3,577 (46.3)	155,469 (50.8)			
High	4,591 (33.6)	2,300 (35.2)	1,996 (32.3)	3,947 (33.5)	576 (34.0)	2,553 (33.0)	79,776 (26.1)			
Missing	643 (4.7)	436 (6.7)	178 (2.9)	569 (4.8)	69 (4.1)	853 (11.0)	15,857 (5.2)			
BMI (kg/m ²)**	25 (24–28)	25 (23– 28)	25 (24– 28)	25 (24–28)	25 (24–27)	25 (24–27)	NA			
Missing before 2010	8,916 (95,3)	4.188 (96.0)	3,792 (94.3)	7,817 (95.6)	983 (93.4)	4,522 (91.9)	NA			
Missing after 2010	2,908 (67.3)	1,526 (70.5)	1,380 (64.1)	2,408 (66.7)	445 (69.2)	1,782 (63.5)	NA			
Hyperlipidaemia/ Lipid-modifying drugs	274 (2.0)	112 (1.7)	145 (2.3)	226 (1.9)	40 (2.4)	129 (1.7)	3,659 (1.2)			
Hypertension/ antihypertensive drugs	882 (6.4)	411 (6.3)	414 (6.7)	756 (6.4)	108 (6.4)	424 (5.5)	14,303 (4.7)			
Birth characteristics										
Sex										
Male	6,775 (49.5)	3,337 (51.1)	2,987 (48.4)	5,868 (49.8)	798 (47.1)	3,923 (50.8)	156,159 (51.0)			
Gestational age, weeks										

(Continued)

	N (%)									
		Chil	dren born a	fter ART	Children born after OI/IUI	Children born after no fertility treatment				
	All ART	cIVF	ICSI	Fresh embryo	Frozen embryo					
<28	89 (0.7)	44 (0.7)	37 (0.6)	79 (0.7)	10 (0.6)	58 (0.8)	566 (0.2)			
28-31	344 (2.5)	189 (2.9)	137 (2.2)	301 (2.6)	35 (2.1)	137 (1.8)	1,906 (0.6)			
32-36	1,283 (9.4)	658 (10.1)	535 (8.7)	1,144 (9.7)	123 (7.3)	570 (7.4)	8,975 (2.9)			
≥37	11,844 (86.6)	5,581 (85.5)	5,414 (87.7)	10,162 (86.2)	1,519 (89.6)	6,896 (89.2)	287,187 (93.9)			
Missing	115 (0.8)	58 (0.9)	52 (0.8)	105 (0.9)	8 (0.5)	67 (0.9)	7,264 (2.4)			
SGA	1,898 (13.9)	946 (14.5)	839 (13.6)	1,702 (14.4)	159 (9.4)	1,032 (13.4)	28,051 (9.2)			
LGA	979 (7.2)	427 (6.5)	449 (7.3)	776 (6.6)	193 (11.4)	621 (8.0)	31,190 (10.2)			
Child from singleton pregnancy	9,914 (72.5)	4,768 (73.0)	4,447 (72.0)	8,515 (72.2)	1,280 (75.5)	6,239 (80.7)	297,562 (97.3)			
Child from multiple pregnancy	3,761 (27.5)	1,762 (37.0)	1,728 (28.0)	3,276 (27.8)	415 (24.4)	1,489 (19.3)	8,336 (2.7)			

Table 2. (Continued)

*Women with BMI above 35 are not treated with ART at public fertility clinics in Denmark.

**Available starting in 2010. Diabetes, hyperlipidaemia/lipid-modifying drugs, and hypertension/antihypertensive drugs are measured before conception. ART, assisted reproductive technologies; BMI, body mass index; cIVF, conventional in vitro fertilization; ICSI, intracytoplasmic sperm Injection; IQR, interquartile range; IUI; intrauterine insemination; IVF, in vitro fertilization; LGA, large for gestational age; NA, not applicable; OI, ovulation induction; SGA, small for gestational age.

https://doi.org/10.1371/journal.pmed.1004324.t002

The associations between fertility treatments and overweight, obesity, and BMI z-scores in children at age 5 to 8 years

After accounting for potential confounding, children born after ART or OI/IUI had the same prevalence of both overweight (11%; adjusted POR 1.00, 95% confidence interval [CI] 0.91 to 1.11; p = 0.95) and obesity (1.9%; adjusted POR 1.01, 95% CI: 0.79 to 1.29; p = 0.94), with minor fluctations of point estimates within subgroups defined by underlying causes of infertility (Fig 2). Differences in mean BMI z-scores between children born after ART versus OI/IUI supported these findings, by yielding a null result (adjusted mean difference of -0.01, 95% CI: -0.05 to 0.02; p = 0.45) (Fig 5).

Comparison of ICSI with conventional IVF yielded adjusted PORs of 0.95 (95% CI: 0.83 to 1.07; p = 0.39) for overweight and 1.16 (95% CI: 0.84 to 1.61; p = 0.36) for obesity (Fig 3), and a difference in the adjusted mean BMI z-score of 0.01 (95% CI: -0.03 to 0.06; p = 0.51) (Fig 5). The association with obesity was modified by underlying causes of infertility, with adjusted PORs ranging from 0.89 (95% CI: 0.49 to 1.62; p = 0.70) for ovulation disorders to 1.41 (95% CI: 0.66 to 2.98; p = 0.37) for the nonspecific female factor category (Fig 3).

Frozen-thawed versus fresh embryo transfer was associated with obesity both before and after accounting for confounding with an adjusted POR of 1.52 (95% CI: 1.09 to 2.17; p = 0.01) (Fig 4). This association remained consistent across all underlying causes of infertility (Fig 4). The mean BMI z-score was higher in children born after frozen-thawed versus fresh embryo transfer [adjusted mean difference = 0.07 (95% CI: 0.01 to 0.13; p = 0.02)] (Fig 5). The frozen-thawed versus fresh embryo association was mainly driven by higher BMI and risk of obesity among children born after frozen-thawed embryo transfer. This was shown by comparing fresh embryo transfer with OI/IUI (adjusted POR of obesity of 0.96, 95% CI: 0.75 to 1.24; p =

	Crude				Adjuste	d				
	Events/ N at risk	Prevalence (%)	e POR (95% CI)	Events/ N at risk	Prevale (%)	nce POF (95%)	R CI)			P-value
All causes			()							
Overweight										
OI/IUI	703/6478	11	Ref	654/6224	11	Ref				
ART	1277/12,219	10	0.92 (0.84-1.01)	1271/12,098	3 11	1.00 (0.91-1.11)	-0-	-		0.95
Obesity										
OI/IUI	131/6478	2.0	Ref	117/6224	1.9	Ref				
ART	230/12,219	1.9	0.95 (0.78-1.16)	233/12,098	1.9	1.01 (0.79-1.29)				0.94
Female factor (any)										
Overweight										
OI/IUI	584/5117	11	Ref	540/4896	11	Ref				
ART	1058/10,070	11	0.85 (0.77-0.94)	1059/9972	11	0.97 (0.86-1.08)		ē		0.54
Obesity										
OI/IUI	105/5117	2.1	Ref	93/4896	1.9	Ref				
ART	199/10,070	2.0	1.00 (0.80-1.24)	204/9972	2.0	1.06 (0.81-1.40)				0.66
Ovulation disorders										
Overweight										
OI/IUI	210/1574	13	Ref	205/1539	13	Ref				
ART	247/2255	11	0.80 (0.66-0.97)	239/2248	11	0.81 (0.67-0.98)				0.03
Obesity										
OI/IUI	46/1574	2.9	Ref	43/1539	2.8	Ref				
ART	55/2255	2.4	0.93 (0.64-1.35)	57/2248	2.5	0.90 (0.59-1.38)				0.63
Tubal factor										
Overweight										
OI/IUI	39/283	14	Ref	37/265	14	Ref				
ART	327/2563	13	0.87 (0.63-1.20)	330/2563	13	0.92 (0.65-1.31)				0.66
Obesity										
OI/IUI	6/283	2.1	Ref	6/265	2.3	Ref				
ART	66/2563	2.6	1.53 (0.67-3.49)	67/2563	2.6	1.12 (0.46-2.75)				0.80
Nonspecific										
Overweight										
OI/IUI	92/859	11	Ref	79/832	9.5	Ref				
ART	220/2331	9.4	0.83 (0.66-1.04)	230/2341	9.8	1.04 (0.81-1.33)				0.78
Obesity										
OI/IUI	21/859	2.4	Ref	16/832	1.9	Ref				
ART	33/2331	1.4	0.64 (0.39-1.05)	35/2341	1.5	0.75 (0.42-1.33)				0.32
Male factor (any)										
Overweight										
OI/IUI	332/3152	11	Ref	298/2939	10	Ref				
ART	923/8768	11	0.98 (0.86-1.11)	942/8779	11	1.06 (0.93-1.20)		F-		0.37
Obesity										
OI/IUI	60/3152	1.9	Ref	53/2939	1.8	Ref				
ART	162/8768	1.8	1.00 (0.76-1.33)	170/8779	1.9	1.07 (0.78-1.45)		—		0.69
									2.0	2.5
							0.0 1.0	/ I.O	2.0	2.0

Fig 2. Overweight and obesity in children born after ARTs compared to OI with or without IUI (OI/IUI), overall and within subgroups defined by underlying causes of infertility. Notes: We adjusted for parental causes of infertility, maternal and paternal age at conception, maternal and paternal highest educational level at conception, maternal country of origin, maternal BMI, maternal smoking status, maternal and paternal hyperlipidemia/use of lipid-modifying drugs, maternal and paternal hypertension/use of antihypertensive drugs, diabetes (type I or II) diagnosed at any time before conception, parity, and year of conception. *P* values were calculated by the large-sample Wald (*Z*) test. Abbreviations: ART, assisted reproductive technology; BMI, body mass index; CI, confidence interval; IUI, intrauterine insemination; OI, ovulation induction; POR, prevalence odds ratio.

https://doi.org/10.1371/journal.pmed.1004324.g002

		Crude			Adjuste	d					
	Events/ N at risk	Prevalence (%)	● POR (95% CI)	Events/ N at risk	Prevalen (%)	ice POF (95%	२ CI)				P-value
All causes		(()			(~ ~				
Overweight											
cIVF	588/5734	10	Ref	614/5672	11	Ref					
ICSI	602/5627	11	1.03 (0.92-1.16)	572/5585	10	0.95 (0.83-1.07)		-			0.39
Obesity											
cIVF	94/5734	1.6	Ref	100/5672	1.8	Ref					
ICSI	112/5627	2.0	1.21 (0.93-1.56)	112/5585	2.0	1.16 (0.84-1.61)	_	-			0.36
Female factor (any)											
Overweight											
cIVF	547/5315	10	Ref	566/5296	11	Ref					
ICSI	432/3993	11	1.00 (0.88-1.14)	406/3938	10	0.97 (0.85-1.11)		_			0.64
Obesity											
clVF	89/5315	1.7	Ref	96/5296	1.8	Ref					
ICSI	90/3993	2.3	1.35 (1.03-1.78)	87/3938	2.2	1.23 (0.88-1.72)	_	-			0.23
Ovulation disorders	S										
Overweight											
cIVF	138/1237	11	Ref	148/1240	12	Ref					
ICSI	90/872	10	0.91 (0.69-1.21)	85/864	9.8	0.83 (0.63-1.09)		-			0.18
Obesity											
cIVF	26/1237	2.1	Ref	31/1240	2.5	Ref					
ICSI	22/872	2.5	1.25 (0.74-2.11)	19/864	2.2	0.89 (0.49-1.62)					0.70
Tubal factor											
Overweight											
clVF	224/1743	13	Ref	225/1731	13	Ref					
ICSI	75/625	12	0.93 (0.71-1.22)	67/614	11	0.83 (0.62-1.11)		_			0.21
Obesity											
clVF	42/1743	2.4	Ref	42/1731	2.4	Ref					
ICSI	14/625	2.2	0.90 (0.51-1.61)	15/614	2.4	0.96 (0.49-1.90)			_		0.91
Nonspecific											
Overweight											
cIVF	126/1369	9.2	Ref	132/1376	9.8	Ref					
ICSI	83/808	10	1.07 (0.82-1.39)	76/804	9.6	0.99 (0.75-1.32)		<u> </u>			0.96
Obesity											
cIVF	15/1369	1.1	Ref	17/1376	1.2	Ref					
ICSI	15/808	1.9	1.57 (0.80-3.05)	14/804	1.7	1.41 (0.66-2.98)					0.37
Male factor (any)											
Overweight											
cIVF	351/3310	11	Ref	364/3277	11	Ref					
ICSI	520/4954	11	0.98 (0.86-1.13)	509/4958	11	0.92 (0.80-1.06)		-			0.26
Obesity											
cIVF	55/3310	1.7	Ref	59/3277	1.8	Ref					
ICSI	91/4954	1.8	1.10 (0.80-1.51)	88/4958	1.8	1.01 (0.69-1.47)		·			0.96
						0).5 1	.0 1.5	2.0	2.5	3.0

Fig 3. Overweight and obesity in children born after ICSI compared to cIVF, overall and within subgroups defined by underlying causes of infertility. Notes: We adjusted for parental causes of infertility, maternal and paternal age at conception, maternal and paternal highest educational level at conception, maternal country of origin, maternal BMI, maternal smoking status, maternal and paternal hyperlipidemia/use of lipid-modifying drugs, maternal and paternal hypertension/use of antihypertensive drugs, diabetes (type I or II) diagnosed at any time before conception, parity, and year of conception. *P* values were calculated by the large-sample Wald (Z) test. Abbreviations: BMI, body mass index; CI, confidence interval; cIVF, conventional in vitro fertilization; ICSI, intracytoplasmic sperm injection; POR, prevalence odds ratio.

https://doi.org/10.1371/journal.pmed.1004324.g003

	Crude				Adjuste	d			
	Events/ N at risk	Prevalenc	e POR (95% CI)	Events/ N at risk	Prevaler (%)	nce POF	R CI)		P-value
All causes		()	()		()	(
Overweight									
Fresh	1099/10,520	10	Ref	1095/10,51	9 10	Ref			
Frozen	169/1553	11	0.96 (0.80-1.14)	172/1539	11	1.08 (0.92-1.26)) –		0.36
Obesity									
Fresh	188/10,520	1.8	Ref	186/10,519	1.8	Ref			
Frozen	42/1553	2.7	1.42 (1.03-1.96)	42/1539	2.7	1.54 (1.09-2.17))		0.01
Female factor (any)									
Overweight									
Fresh	901/8625	10	Ref	899/8624	10	Ref			
Frozen	147/1299	11	0.99 (0.82-1.20)	148/1285	12	1.10 (0.93-1.31))		0.26
Obesity									
Fresh	162/8625	1.9	Ref	161/8624	1.9	Ref			
Frozen	37/1299	2.8	1.42 (1.01-2.00)	36/1285	2.8	1.47 (1.02-2.13))		0.04
Ovulation disorders									
Overweight									
Fresh	206/1934	11	Ref	205/1938	11	Ref			
Frozen	34/241	14	1.24 (0.84-1.81)	35/236	15	1.38 (0.97-1.97)) .		0.07
Obesity									
Fresh	45/1934	2.3	Ref	44/1938	2.3	Ref			
Frozen	9/241	3.7	1.46 (0.75-2.87)	10/236	4.2	1.73 (0.97-3.68))	-	0.14
Tubal factor									
Overweight									
Fresh	268/2146	12	Ref	267/2148	12	Ref			
Frozen	53/377	14	1.01 (0.73-1.38)	55/372	15	1.18 (0.89-1.57))		0.26
Obesity									
Fresh	49/2146	2.3	Ref	49/2148	2.3	Ref			
Frozen	16/377	4.2	1.76 (1.05-2.97)	15/372	4.0	1.73 (0.97-3.06))	-	- 0.06
Nonspecific									
Overweight									
Fresh	181/1953	9.3	Ref	182/1960	9.3	Ref			
Frozen	30/295	10	0.76 (0.48-1.18)	28/285	9.8	1.07 (0.72-1.60))		0.73
Obesity									
Fresh	23/1953	1.2	Ref	23/1960	1.2	Ref			
Frozen	10/295	3.4	2.49 (1.23-5.05)	10/285	3.5	2.89 (1.29-6.43)) – –		0.01
Male factor (any)									
Overweight									
Fresh	806/7634	11	Ref	805/7633	11	Ref			
Frozen	115/1083	11	0.93 (0.76-1.14)	116/1068	11	1.04 (0.86-1.26))		0.70
Obesity									
Fresh	135/7634	1.8	Ref	135/7633	1.8	Ref			
Frozen	27/1083	2.5	1.29 (0.86-1.93)	27/1068	2.5	1.45 (0.94-2.23))		0.09
							0.5 1.0 1.5	2.0 2.5 3.	0 3.5

Fig 4. Overweight and obesity in children born after frozen-thawed embryo transfer compared to fresh embryo transfer, overall and within subgroups defined by underlying causes of infertility. Notes: We adjusted for parental causes of infertility, maternal and paternal age at conception, maternal and paternal highest educational level at conception, maternal country of origin, maternal BMI, maternal smoking status, maternal and paternal hyperlipidemia/use of lipid-modifying drugs, maternal and paternal hypertension/use of antihypertensive drugs, diabetes (type I or II) diagnosed at any time before conception, parity, and year of conception. Distribution of fresh vs. frozen embryo transfer by previous number of ART cycles: 0 cycles: 97.2% vs. 2.75%, 1 cycle: 78.9% vs. 21.1%, 2 cycles: 83.8% vs. 16.2%, ≥3 cycles: 84.5% vs. 15.5%. Distribution of fresh vs. frozen embryo transfer by maternal age (years): <25: 88.6% vs. 11.4%, 25-

29: 87.9% vs. 12.1%, 30-34: 87.2% vs. 12.7%, 35-39: 87.1% vs. 12.9%, \geq 40: 88.3% vs. 11.7%. *P* values were calculated by the large-sample Wald (Z) test. Abbreviations: ART, assisted reproductive technology; BMI, body mass index; CI, confidence interval; POR, prevalence odds ratio.

https://doi.org/10.1371/journal.pmed.1004324.g004

0.78) and frozen-thawed embryo transfer with OI/IUI (adjusted POR of obesity of 1.44, 95% CI: 0.96 to 2.17; p = 0.08) (S2, S3, and S4 Figs), respectively.

Sex of the child had a slight impact on the estimates, with a tendency towards higher point estimates in boys than in girls (<u>S5 Text</u>). Sensitivity analyses produced minor fluctuations in



Fig 5. Crude and adjusted differences in mean BMI Z-scores, overall and within subgroups defined by underlying causes of infertility. For children born after: (A) ARTs compared to OI with or without IUI (OI/IUI), (B) ICSI compared to cIVF, and (C) frozen-thawed embryo transfer compared to fresh embryo transfer. Notes: We adjusted for parental causes of infertility, maternal and paternal age at conception, maternal and paternal highest educational level at conception, maternal country of origin, maternal BMI, maternal smoking status, maternal and paternal hyperlipidemia/use of lipid-modifying drugs, maternal and paternal hypertension/use of antihypertensive drugs, diabetes (type I or II) diagnosed at any time before conception, parity, and year of conception. *P* values were calculated using the large-sample Wald (Z) test. Abbreviation: ART, assisted reproductive technology; BMI, body mass index; CI, confidence interval; cIVF, conventional in vitro fertilization; IUI, intrauterine insemination; OI, ovulation induction.

https://doi.org/10.1371/journal.pmed.1004324.g005

estimates without consequences for our overall interpretation (S6 Text). Frozen-thawed embryo transfer in modified natural cycle (using hCG trigger only) and programmed cycle (estrogen and progesterone) yielded similar results (S6 Text). The supplemental analyses, using the general population comparator, provided similar direction of estimates as the main analyses (S7 and S8 Texts).

Discussion

This nationwide population-based cohort study found no association with BMI at age 5 to 8 years when comparing ART versus OI/IUI and ICSI versus conventional IVF. However, use of frozen-thawed embryo transfer was associated with an excess risk of obesity compared to fresh embryo transfer and OI/IUI, respectively. Most associations appeared to be only modestly influenced by the underlying cause of infertility (no evidence of substantial effect modification).

This study expands prior knowledge in multiple ways. Importantly, the reasons that couples seek or receive fertility treatments could impact child outcomes, i.e., the indication for treatment itself could be influential. To address this concern, we only compared children born to parents receiving some form of fertility treatment. Hence, we reduced confounding from parental health, infertility, and ability/willingness to access fertility treatment. Moreover, the large study size, population approach, and use of medical registries with detailed clinical information allowed us to perform analyses by specific ART procedures and underlying causes of infertility. To provide information comparable to those of some prior studies, we also described outcomes among children born after fertility treatment and among children born after no fertility treatment, finding the first group at lower risk of obesity at age 5 to 8 years. This is consistent with results from 2 recent large cohort studies conducted by Elhakeem and colleagues (N = 158,066, of which 4,329 were born after ART) and Magnus and colleagues (N = 81,461, of which 1,721 were born after ART) [20,21].

We found a consistent positive association between frozen-thawed embryo transfer and obesity. Similarly, Magnus and colleagues reported higher mean BMI Z-scores for children born after frozen-thawed versus fresh embryo transfer up to age 6, but no difference at age 17 [21]. Hann and colleagues reported an adjusted mean BMI difference of 0.043 kg/m² (95% CI: -0.150 to 0.233) at age 5 year for children born after frozen-thawed (n = 1,091) versus fresh embryo transfer (n = 4,127) [6]. While frozen-thawed embryo transfer was associated with an elevated risk of obesity compared with fresh embryo transfer, we found that the crude risk was close to that observed in children from the general population born after no fertility treatment. This result was supported by Elhakeem and colleagues and Magnus and colleagues [20,21]. Former studies reported different obstetric and perinatal outcomes depending on the specific IVF protocols. As example, Busnelli and colleagues and Versen-Höynck and colleagues found higher risk of hypertensive disorders of pregnancy, preeclampsia, postpartum haemorrhage, and cesarean section when comparing frozen-thawed embryo transfer in programmed versus natural cycle [47,48]. In our study, we found similar associations for both types of endometrial preparation.

For ICSI, we confirmed the null association observed in prior studies [3,6,10,11,13,16,21]. Our observation of a tendency towards higher point estimates in boys than in girls is somewhat unsupported in the literature, but existing evidence is sparse [14,20,21]. Still, sex-specific differences in metabolic programming is widely acknowledged [46,49].

Our study has several potential limitations. First, approximately 13% of eligible children lacked outcome measurements. Recording of weight and height in the Danish National Children's Database comes from routine examinations of all Danish children, but missing

information occurs due to technical issues, such as randomly occurring local breakdowns of reporting systems, or removal of implausible measurements as part of a central validation process. Importantly, we found that the occurrence of missing BMI measurement was similar across all exposure and comparison groups. In addition, our evaluation of the data indicated that censoring would be noninformative. Second, our findings should be interpreted in light of restriction to live born children. If our exposures affected the probability of live birth, this restriction could lead to live birth bias in the presence of unmeasured common causes of fetal loss and high BMI [50]. It is possible that the probability of having a live birth differed across the exposure and comparison cohorts. Because of lack of data not resulting in livebirths, we were unable to examine this issue. Third, with respect to confounding, we used a comparison cohort of children born to parents receiving fertility treatment. Characteristics of parents in the ART and OI/IUI cohorts were homogenous with respect to measured confounders and plausibly also were somewhat alike with respect to unmeasured or unknown confounders. Still, we do not have data on the unmeasured or unknown, thus cannot rule out confounding. One example of a potentially important poorly measured factor is the severity of infertility. Fourth, we cannot dismiss nondifferential misclassification of treatments, outcomes, or covariates. Danish children born following fertility treatment abroad (i.e., reproductive tourism) would be misclassified as being born after no fertility treatment. Further, children born after no fertility treatment could have parents with infertility. The Danish registries do not capture time to pregnancy, but 5.4% of children in this group had parents with a prior diagnosis of infertility. Some children in the cohort were born after fertility treatment using donor oocytes or sperm. For these, the listed baseline characteristics concerns the registered parents, not the biological donor. While this recording should be appropriate for some characteristics (i.e., parental age, educational level, and smoking), it might be less appropriate for other traits or diagnoses with a genetic component. Fifth, we lacked detailed technical information about specific procedures, such as those used for freezing and thawing. Finally, our study did not explore the risk of obesity in adolescence or adulthood, or the development of metabolic diseases. Body fat composition and other elements of the metabolic syndrome are essential to consider [20,21].

Our overall null results provide reassuring results for couples with infertility seeking help. Our finding of elevated obesity risk after frozen-thawed versus fresh embryo transfer could reflect multiple pathways. First, gametes and embryos are exposed to hormones, in vitro manipulation, and culturing in artificial media during ART [7]. Such stimuli may lead to epigenetic programming of the metabolism of children, with a net result of increased obesity risk in a later obesogenic environment [51,52]. The potential impact of epigenetic programming may differ in frozen-thawed and fresh embryos due to differences in the in vitro manipulation techniques, and hormonal and intrauterine environments [51]. Also, the process of freezing and thawing might induce molecular changes in the embryo. Second, accumulating evidence consistently links the use of frozen-thawed embryos to larger weights at birth, which may persist into later life. Intrauterine growth conditions could be affected by various mechanisms, such as placental function, growth signalling pathways, or pregnancy complications (i.e., gestational diabetes). Third, couples with certain unknown or unmeasured traits or characteristics could be more likely to progress from fresh to frozen-thawed embryo treatment. Hence, confounding from such background factors is another possible explanation to the findings. Finally, given the multiple comparisons, our result may be a chance finding and thus needs confirmation in future studies.

In conclusion, our study found no evidence supporting an association between ART and BMI at age 5 to 8 years in a population of children born to parents treated for infertility. However, longer follow-up and assessment of other adverse metabolic outcomes need consideration. Use of frozen-thawed embryo transfer was associated with an excess risk of obesity compared to fresh embryo transfer and OI/IUI, respectively. Despite elevated relative risks, the absolute risk differences were low. Reasons that underly this finding requires more investigation.

Supporting information

S1 STROBE Checklist. STROBE Statement—Checklist of items that should be included in reports of *cohort studies*.

(PDF)

S1 Text. Original analysis plan. (PDF)

S2 Text. Characteristics of children with and without routine anthropometric evaluation at age 5–8 years.

(PDF)

S3 Text. Definition of exposure and covariates. (PDF)

S4 Text. Crude and adjusted prevalence odds ratios (PORs) between covariates and overweight and obesity.

```
(PDF)
```

S5 Text. Overweight and obesity in children born after assisted reproductive technologies, stratified by sex of the child. (PDF)

S6 Text. Results from sensitivity analyses.

(PDF)

S7 Text. Crude and adjusted prevalences (%) and prevalence odds ratios (PORs) for overweight and obesity in children born after assisted reproductive technologies (ARTs) vs. children from the general population. (PDF)

S8 Text. Crude and adjusted differences in mean BMI Z-scores comparing children born after assisted reproductive technologies (ARTs) vs. children from the general population. (PDF)

S1 Fig. Propensity score distributions across exposure and comparison cohorts overall and for subgroups defined by underlying cause of infertility for (A) ARTs vs. OI with or without IUI (OI/IUI), (B) ICSI vs. cIVF, and (C) frozen-thawed vs. fresh embryo transfer. Notes: Abbreviations: ART, assisted reproductive technology; cIVF, conventional in vitro fertilization; ICSI, intracytoplasmic sperm injection; IUI, intrauterine insemination; OI; ovulation induction.

(PDF)

S2 Fig. Overweight and obesity in children born after fresh embryo transfer compared to OI with or without IUI (OI/IUI), overall and within subgroups defined by underlying causes of infertility. Notes: We adjusted for parental causes of infertility, maternal and paternal age at conception, maternal and paternal highest educational level at conception, maternal country of origin, maternal BMI, maternal smoking status, maternal and paternal hyperlipidemia/use of lipid-modifying drugs, maternal and paternal hypertension/use of antihypertensive drugs, diabetes (type I or II) diagnosed at any time before conception, parity, and year of conception. *P* values were calculated by the large-sample Wald (Z) test. Abbreviations: BMI, body mass index; CI, confidence interval; IUI, intrauterine insemination; OI, ovulation induction; POR, prevalence odds ratio.

(PDF)

S3 Fig. Overweight and obesity in children born after frozen-thawed embryo transfer compared to OI with or without IUI (OI/IUI), overall and within subgroups defined by underlying causes of infertility. Notes: We adjusted for parental causes of infertility, maternal and paternal age at conception, maternal and paternal highest educational level at conception, maternal country of origin, maternal BMI, maternal smoking status, maternal and paternal hyperlipidemia/use of lipid-modifying drugs, maternal and paternal hypertension/use of antihypertensive drugs, diabetes (type I or II) diagnosed at any time before conception, parity, and year of conception. All standardized differences were ≤ 0.15 after weighting, except for maternal BMI, which yielded an standardized difference of 0.29. *P* values were calculated using the large-sample Wald (*Z*) test. Abbreviations: BMI, body mass index; CI, confidence interval; IUI, intrauterine insemination; OI, ovulation induction; POR, prevalence odds ratio. (PDF)

S4 Fig. Crude and adjusted differences in mean BMI Z-scores, overall and within subgroups defined by underlying causes of infertility. For children born after: (A) fresh embryo transfer compared to OI with or without IUI (OI/IUI) and (B) frozen-thawed embryo transfer compared to OI/IUI. Notes: We adjusted for parental causes of infertility, maternal and paternal age at conception, maternal and paternal highest educational level at conception, maternal country of origin, maternal BMI, maternal smoking status, maternal and paternal hyperlipidemia/use of lipid-modifying drugs, maternal and paternal hypertension/use of antihypertensive drugs, diabetes (type I or II) diagnosed at any time before conception, parity, and year of conception. All standardized differences were ≤ 0.15 after weighting, except for maternal BMI, which yielded an standardized difference of 0.29. *P* values were calculated by the large-sample Wald (Z) test. Abbreviation: BMI, body mass index; CI, confidence interval; IUI, intrauterine insemination; OI, ovulation induction. (PDF)

Author Contributions

Conceptualization: Kristina Laugesen, Katalin Veres, Sonia Hernandez-Diaz, Yu-Han Chiu, Anna Sara Oberg, John Hsu, Paolo Rinaudo, Mandy Spaan, Flora van Leeuwen, Henrik Toft Sørensen.

Data curation: Kristina Laugesen, Katalin Veres, Henrik Toft Sørensen.

Formal analysis: Katalin Veres.

Funding acquisition: Sonia Hernandez-Diaz, Yu-Han Chiu, John Hsu.

- **Investigation:** Kristina Laugesen, Sonia Hernandez-Diaz, Yu-Han Chiu, Anna Sara Oberg, John Hsu, Paolo Rinaudo, Mandy Spaan, Flora van Leeuwen, Henrik Toft Sørensen.
- **Methodology:** Kristina Laugesen, Katalin Veres, Sonia Hernandez-Diaz, Yu-Han Chiu, Anna Sara Oberg, John Hsu, Paolo Rinaudo, Mandy Spaan, Flora van Leeuwen, Henrik Toft Sørensen.
- Project administration: Kristina Laugesen, Henrik Toft Sørensen.

- Supervision: Katalin Veres, Sonia Hernandez-Diaz, Yu-Han Chiu, Anna Sara Oberg, John Hsu, Paolo Rinaudo, Mandy Spaan, Henrik Toft Sørensen.
- Validation: Katalin Veres, Sonia Hernandez-Diaz, Yu-Han Chiu, Anna Sara Oberg, John Hsu, Paolo Rinaudo, Mandy Spaan, Henrik Toft Sørensen.
- Visualization: Kristina Laugesen, Katalin Veres.
- Writing original draft: Kristina Laugesen, Sonia Hernandez-Diaz, Yu-Han Chiu, Anna Sara Oberg, John Hsu, Paolo Rinaudo, Mandy Spaan, Henrik Toft Sørensen.
- Writing review & editing: Kristina Laugesen, Katalin Veres, Sonia Hernandez-Diaz, Yu-Han Chiu, Anna Sara Oberg, John Hsu, Paolo Rinaudo, Mandy Spaan, Flora van Leeuwen, Henrik Toft Sørensen.

References

- European IVF-Monitoring Consortium (EIM) for the European Society of Human Reproduction and Embryology (ESHRE), Wyns C, De Geyter C, Calhaz-Jorge C, Kupka MS, et al. ART in Europe, 2017: results generated from European registries by ESHRE†. Hum Reprod Open. 2021; 2021(3). https://doi. org/10.1093/hropen/hoab026 PMID: 34377841
- 2. The World Health Organization. Infertility is a global public health issue [cited 2022 Mar 1]. Available from: https://www.who.int/reproductivehealth/topics/infertility/perspective/en/.
- Cui L, Zhou W, Xi B, Ma J, Hu J, Fang M, et al. Increased risk of metabolic dysfunction in children conceived by assisted reproductive technology. Diabetologia. 2020. <u>https://doi.org/10.1007/s00125-020-05241-1 PMID: 32757153</u>
- Ceelen M, van Weissenbruch MM, Roos JC, Vermeiden JP, van Leeuwen FE, Delemarre-van de Waal HA. Body composition in children and adolescents born after in vitro fertilization or spontaneous conception. J Clin Endocrinol Metab. 2007; 92(9):3417–3423. Epub 2007/06/28. https://doi.org/10.1210/jc. 2006-2896 PMID: 17595253.
- Ceelen M, van Weissenbruch MM, Vermeiden JP, van Leeuwen FE, Delemarre-van de Waal HA. Cardiometabolic differences in children born after in vitro fertilization: follow-up study. J Clin Endocrinol Metab. 2008; 93(5):1682–1688. Epub 2008/02/21. https://doi.org/10.1210/jc.2007-2432 PMID: 18285409.
- Hann M, Roberts SA, D'Souza SW, Clayton P, Macklon N, Brison DR. The growth of assisted reproductive treatment-conceived children from birth to 5 years: a national cohort study. BMC Med. 2018; 16 (1):224. Epub 2018/11/30. <u>https://doi.org/10.1186/s12916-018-1203-7</u> PMID: <u>30482203</u>; PubMed Central PMCID: PMC6260690.
- Hart R, Norman RJ. The longer-term health outcomes for children born as a result of IVF treatment: Part I—General health outcomes. Hum Reprod Update. 2013; 19(3):232–243. Epub 2013/03/02. https://doi.org/10.1093/humupd/dms062 PMID: 23449642.
- Chen M, Heilbronn LK. The health outcomes of human offspring conceived by assisted reproductive technologies (ART). J Dev Orig Health Dis. 2017; 8(4):388–402. Epub 2017/04/19. <u>https://doi.org/10.1017/S2040174417000228</u> PMID: 28416029.
- Norrman E, Petzold M, Gissler M, Spangmose AL, Opdahl S, Henningsen AK, et al. Cardiovascular disease, obesity, and type 2 diabetes in children born after assisted reproductive technology: A population-based cohort study. PLoS Med. 2021; 18(9):e1003723. Epub 2021/09/08. https://doi.org/10.1371/journal.pmed.1003723 PMID: 34491995; PubMed Central PMCID: PMC8423242.
- Kai CM, Main KM, Andersen AN, Loft A, Chellakooty M, Skakkebaek NE, et al. Serum insulin-like growth factor-I (IGF-I) and growth in children born after assisted reproduction. J Clin Endocrinol Metab. 2006; 91(11):4352–4360. Epub 2006/08/17. https://doi.org/10.1210/jc.2006-0701 PMID: 16912121.
- Knoester M, Helmerhorst FM, Vandenbroucke JP, van der Westerlaken LAJ, Walther FJ, Veen S. Perinatal outcome, health, growth, and medical care utilization of 5- to 8-year-old intracytoplasmic sperm injection singletons. Fertil Steril. 2008; 89(5):1133–1146. Epub 2008/01/08. https://doi.org/10.1016/j. fertnstert.2007.04.049 PMID: 18177652.
- Sakka SD, Loutradis D, Kanaka-Gantenbein C, Margeli A, Papastamataki M, Papassotiriou I, et al. Absence of insulin resistance and low-grade inflammation despite early metabolic syndrome manifestations in children born after in vitro fertilization. Fertil Steril. 2010; 94(5):1693–1699. Epub 2010/01/05. https://doi.org/10.1016/j.fertnstert.2009.09.049 PMID: 20045517.

- Ludwig AK, Katalinic A, Thyen U, Sutcliffe AG, Diedrich K, Ludwig M. Physical health at 5.5 years of age of term-born singletons after intracytoplasmic sperm injection: results of a prospective, controlled, single-blinded study. Fertil Steril. 2009; 91(1):115–124. Epub 2008/01/22. https://doi.org/10.1016/j. fertnstert.2007.11.037 PMID: 18206144.
- Belva F, Painter R, Bonduelle M, Roelants M, Devroey P, De Schepper J. Are ICSI adolescents at risk for increased adiposity? Hum Reprod. 2012; 27(1):257–264. Epub 2011/11/15. https://doi.org/10.1093/ humrep/der375 PMID: 22081314.
- Green MP, Mouat F, Miles HL, Hopkins SA, Derraik JG, Hofman PL, et al. Phenotypic differences in children conceived from fresh and thawed embryos in in vitro fertilization compared with naturally conceived children. Fertil Steril. 2013; 99(7):1898–1904. Epub 2013/03/12. https://doi.org/10.1016/j. fertnstert.2013.02.009 PMID: 23472944.
- Gkourogianni A, Kosteria I, Telonis AG, Margeli A, Mantzou E, Konsta M, et al. Plasma metabolomic profiling suggests early indications for predisposition to latent insulin resistance in children conceived by ICSI. PLoS ONE. 2014; 9(4):e94001. Epub 2014/04/15. https://doi.org/10.1371/journal.pone.0094001 PMID: 24728198; PubMed Central PMCID: PMC3984097.
- Pontesilli M, Painter RC, Grooten IJ, van der Post JA, Mol BW, Vrijkotte TG, et al. Subfertility and assisted reproduction techniques are associated with poorer cardiometabolic profiles in childhood. Reprod Biomed Online. 2015; 30(3):258–267. Epub 2015/01/17. <u>https://doi.org/10.1016/j.rbmo.2014.11.006</u> PMID: 25592973.
- Guo XY, Liu XM, Jin L, Wang TT, Ullah K, Sheng JZ, et al. Cardiovascular and metabolic profiles of offspring conceived by assisted reproductive technologies: a systematic review and meta-analysis. Fertil Steril. 2017; 107(3):622–631.e5. Epub 2017/01/21. https://doi.org/10.1016/j.fertnstert.2016.12.007 PMID: 28104241.
- Cui L, Zhao M, Zhang Z, Zhou W, Lv J, Hu J, et al. Assessment of Cardiovascular Health of Children Ages 6 to 10 Years Conceived by Assisted Reproductive Technology. JAMA Netw Open. 2021; 4(11): e2132602. Epub 2021/11/05. https://doi.org/10.1001/jamanetworkopen.2021.32602 PMID: 34735014; PubMed Central PMCID: PMC8569486.
- Elhakeem A, Taylor AE, Inskip HM, Huang J, Tafflet M, Vinther JL, et al. Association of Assisted Reproductive Technology With Offspring Growth and Adiposity From Infancy to Early Adulthood. JAMA Netw Open. 2022; 5(7):e2222106. Epub 2022/07/27. <u>https://doi.org/10.1001/jamanetworkopen.2022.22106</u> PMID: 35881399.
- Magnus MC, Wilcox AJ, Fadum EA, Gjessing HK, Opdahl S, Juliusson PB, et al. Growth in children conceived by ART. Hum Reprod. 2021; 36(4):1074–1082. Epub 2021/02/17. https://doi.org/10.1093/ humrep/deab007 PMID: 33592626; PubMed Central PMCID: PMC7970724.
- Pinborg A, Loft A, Aaris Henningsen AK, Rasmussen S, Andersen AN. Infant outcome of 957 singletons born after frozen embryo replacement: the Danish National Cohort Study 1995–2006. Fertil Steril. 2010; 94(4):1320–1327. Epub 2009/08/04. https://doi.org/10.1016/j.fertnstert.2009.05.091 PMID: 19647236.
- Pelkonen S, Koivunen R, Gissler M, Nuojua-Huttunen S, Suikkari AM, Hydén-Granskog C, et al. Perinatal outcome of children born after frozen and fresh embryo transfer: the Finnish cohort study 1995–2006. Hum Reprod. 2010; 25(4):914–923. Epub 2010/02/04. https://doi.org/10.1093/humrep/dep477 PMID: 20124395.
- Laugesen K, Ludvigsson JF, Schmidt M, Gissler M, Valdimarsdottir UA, Lunde A, et al. Nordic Health Registry-Based Research: A Review of Health Care Systems and Key Registries. Clin Epidemiol. 2021; 13:533–554. Epub 2021/07/30. <u>https://doi.org/10.2147/CLEP.S314959</u> PMID: <u>34321928</u>; PubMed Central PMCID: PMC8302231.
- Blenstrup LT, Knudsen LB. Danish registers on aspects of reproduction. Scand J Public Health. 2011; 39(7 Suppl):79–82. Epub 2011/08/04. https://doi.org/10.1177/1403494811399957 PMID: 21775359.
- Rambøll Management Consulting. Undersøgelse af hjemmeundervisning, fravær, og børn uden for undervisningstilbud. 2018. Available from: https://www.uvm.dk/publikationer/folkeskolen/2018undersoegelse-af-hjemmeundervisning—fravaer-og-boern-uden-for-undervisningstilbud.
- 27. Schmidt M, Schmidt SA, Sandegaard JL, Ehrenstein V, Pedersen L, Sorensen HT. The Danish National Patient Registry: a review of content, data quality, and research potential. Clin Epidemiol. 2015; 7:449–490. https://doi.org/10.2147/CLEP.S91125 PMID: 26604824
- Pottegard A, Schmidt SA, Wallach-Kildemoes H, Sorensen HT, Hallas J, Schmidt M. Data Resource Profile: The Danish National Prescription Registry. Int J Epidemiol. 2017; 46(3):798–798f. https://doi. org/10.1093/ije/dyw213 PMID: 27789670; PMCID: PMC5837522.
- 29. Schmidt M, Pedersen L, Sorensen HT. The Danish Civil Registration System as a tool in epidemiology. Eur J Epidemiol. 2014; 29(8):541–549. https://doi.org/10.1007/s10654-014-9930-3 PMID: 24965263
- Knudsen LB, Olsen J. The Danish Medical Birth Registry. Dan Med Bull. 1998; 45(3):320–323. PMID: 9675544

- WorldObesity. Obesity Classification [cited 2020 Jun 12]. Available from: https://www.worldobesity.org/about/about-obesity/obesity-classification.
- Cole TJ, Bellizzi MC, Flegal KM, Dietz WH. Establishing a standard definition for child overweight and obesity worldwide: international survey. BMJ. 2000; 320(7244):1240–1243. Epub 2000/05/08. https:// doi.org/10.1136/bmj.320.7244.1240 PMID: 10797032; PubMed Central PMCID: PMC27365.
- WHO. Growth reference data for 5–19 years [cited 2022 Jun 21]. Available from: https://www.who.int/ toolkits/growth-reference-data-for-5to19-years/application-tools.
- Chen X, Koivuaho E, Piltonen TT, Gissler M, Lavebratt C. Association of maternal polycystic ovary syndrome or anovulatory infertility with obesity and diabetes in offspring: a population-based cohort study. Hum Reprod. 2021; 36(8):2345–2357. Epub 2021/05/29. https://doi.org/10.1093/humrep/deab112 PMID: 34046665; PubMed Central PMCID: PMC8289324.
- Barclay K, Myrskylä M. Maternal age and offspring health and health behaviours in late adolescence in Sweden. SSM Popul Health. 2016; 2:68–76. <u>https://doi.org/10.1016/j.ssmph.2016.02.012</u> PMID: 28470034
- Barriuso L, Miqueleiz E, Albaladejo R, Villanueva R, Santos JM, Regidor E. Socioeconomic position and childhood-adolescent weight status in rich countries: a systematic review, 1990–2013. BMC Pediatr. 2015; 15:129. Epub 2015/09/24. <u>https://doi.org/10.1186/s12887-015-0443-3</u> PMID: <u>26391227</u>; PubMed Central PMCID: PMC4578240.
- Pedersen DC, Aarestrup J, Pearson S, Baker JL. Ethnic Inequalities in Overweight and Obesity Prevalence among Copenhagen Schoolchildren from 2002 to 2007. Obes Facts. 2016; 9(4):284–295. Epub 2016/09/01. https://doi.org/10.1159/000446482 PMID: 27577690; PubMed Central PMCID: PMC5644825.
- Voerman E, Santos S, Patro Golab B, Amiano P, Ballester F, Barros H, et al. Maternal body mass index, gestational weight gain, and the risk of overweight and obesity across childhood: An individual participant data meta-analysis. PLoS Med. 2019; 16(2):e1002744. Epub 2019/02/12. https://doi.org/10. 1371/journal.pmed.1002744 PMID: 30742624; PubMed Central PMCID: PMC6370184
- Schnurr TM, Ängquist L, Nøhr EA, Hansen T, Sørensen TIA, Morgen CS. Smoking during pregnancy is associated with child overweight independent of maternal pre-pregnancy BMI and genetic predisposition to adiposity. Sci Rep. 2022; 12(1):3135. Epub 2022/02/26. https://doi.org/10.1038/s41598-022-07122-6 PMID: 35210505; PubMed Central PMCID: PMC8873398.
- 40. Øyri LKL, Christensen JJ, Sebert S, Thoresen M, Michelsen TM, Ulven SM, et al. Maternal prenatal cholesterol levels predict offspring weight trajectories during childhood in the Norwegian Mother, Father and Child Cohort Study. BMC Med. 2023; 21(1):43. <u>https://doi.org/10.1186/s12916-023-02742-9</u> PMID: 36747215
- Li R, Alpert BS, Walker SS, Somes GW. Longitudinal Relationship of Parental Hypertension with Body Mass Index, Blood Pressure, and Cardiovascular Reactivity in Children. J Pediatr. 2007; 150(5):498– 502. https://doi.org/10.1016/j.jpeds.2007.01.034 PMID: 17452224
- 42. Pitchika A, Jolink M, Winkler C, Hummel S, Hummel N, Krumsiek J, et al. Associations of maternal type 1 diabetes with childhood adiposity and metabolic health in the offspring: a prospective cohort study. Diabetologia. 2018; 61(11):2319–2332. https://doi.org/10.1007/s00125-018-4688-x PMID: 30008062
- **43.** Gaillard R, Rurangirwa AA, Williams MA, Hofman A, Mackenbach JP, Franco OH, et al. Maternal parity, fetal and childhood growth, and cardiometabolic risk factors. Hypertension. 2014; 64(2):266–274. Epub 2014/05/29. https://doi.org/10.1161/HYPERTENSIONAHA.114.03492 PMID: 24866145.
- Schramm S, Sørensen TIA, Davidsen M, Tolstrup JS. Changes in adult obesity prevalence in Denmark, 1987–2021: age–period–cohort analysis of nationally representative data. Eur J Public Health. 2023; 33 (3):463–467. <u>https://doi.org/10.1093/eurpub/ckad024</u> J European Journal of Public Health PMID: 36840661
- Desai RJ, Franklin JM. Alternative approaches for confounding adjustment in observational studies using weighting based on the propensity score: a primer for practitioners. BMJ. 2019; 367:I5657. Epub 2019/10/28. https://doi.org/10.1136/bmj.I5657 PMID: 31645336.
- Dearden L, Bouret SG, Ozanne SE. Sex and gender differences in developmental programming of metabolism. Mol Metab. 2018; 15:8–19. Epub 2018/05/19. https://doi.org/10.1016/j.molmet.2018.04. 007 PMID: 29773464; PubMed Central PMCID: PMC6066743.
- 47. Busnelli A, Schirripa I, Fedele F, Bulfoni A, Levi-Setti PE. Obstetric and perinatal outcomes following programmed compared to natural frozen-thawed embryo transfer cycles: a systematic review and meta-analysis. Hum Reprod. 2022; 37(7):1619–1641. https://doi.org/10.1093/humrep/deac073 J Human Reproduction PMID: 35553678
- 48. von Versen-Höynck F, Schaub AM, Chi YY, Chiu KH, Liu J, Lingis M, et al. Increased Preeclampsia Risk and Reduced Aortic Compliance With In Vitro Fertilization Cycles in the Absence of a Corpus

Luteum. Hypertension. 2019; 73(3):640–649. Epub 2019/01/15. https://doi.org/10.1161/ HYPERTENSIONAHA.118.12043 PMID: 30636552; PubMed Central PMCID: PMC6434532.

- Laugesen K, Sørensen HT, Jorgensen JOL, Petersen I. Prenatal exposure to glucocorticoids and the prevalence of overweight or obesity in childhood. Eur J Endocrinol. 2022; 186(4):429–440. Epub 2022/ 02/02. https://doi.org/10.1530/EJE-21-0846 PMID: 35104239; PubMed Central PMCID: PMC8942335.
- 50. Liew Z, Olsen J, Cui X, Ritz B, Arah OA. Bias from conditioning on live birth in pregnancy cohorts: an illustration based on neurodevelopment in children after prenatal exposure to organic pollutants. Int J Epidemiol. 2015; 44(1):345–354. Epub 2015/01/22. https://doi.org/10.1093/ije/dyu249 PMID: 25604449; PubMed Central PMCID: PMC4339763.
- Cannarella R, Crafa A, Mongioì LM, Leggio L, Iraci N, La Vignera S, et al. DNA Methylation in Offspring Conceived after Assisted Reproductive Techniques: A Systematic Review and Meta-Analysis. J Clin Med. 2022; 11(17). Epub 2022/09/10. https://doi.org/10.3390/jcm11175056 PMID: 36078985.
- Carlsen E, Lee Y, Magnus P, Jugessur A, Page CM, Nustad HE, et al. An examination of mediation by DNA methylation on birthweight differences induced by assisted reproductive technologies. Clin Epigenetics. 2022; 14(1):151. Epub 2022/11/30. https://doi.org/10.1186/s13148-022-01381-w PMID: 36443807.